Author's response to reviews

Title: Assessment of the health care waste generation rate and its management system in Hospitals in Addis Ababa, Ethiopia, 2011.

Authors:

Mesfin K Debere Mr. (messi.kid@gmail.com)

Version: 2 Date: 22 July 2012

Author's response to reviews:

This is the revised Manuscript according to the BMC guideline.